MATHEMATICS 300 — SPRING 1997 Introduction to Mathematical Reasoning

Instructor: H. J. Sussmann HANDOUT NO. 15. April 29, 1997

The FINAL EXAM will be on Thursday, May 8, from 8:00 p.m. to 11:00 p.m., in room CAC HH-A6 (i.e. the room where our classes are held).

READING ASSIGNMENT FOR APRIL 29 AND MAY 1: Chapter 8, Sections 1, 2 and 3, with special emphasis on the discussion of completeness. (See Handouts 1 to 14 for previous reading assigments.)

HOMEWORK ASSIGNMENT NO. 12, DUE ON THURSDAY, MAY 1: Pages 178-9, Problems 11 and 14. Page 191-2, Problems 2 and 8. Page 309, Problems 1 and 8. (See previous handouts for Assigments 1 to 11.)